Author’s response to reviews

Title: Cognitive appraisal of exposure to specific types of trauma. A study of gender differences

Authors:

Dana HERTA (dana_cristina75@yahoo.com)
Doina COZMAN (doina_octaviancosman@yahoo.com)

Version: 3 Date: 04 Aug 2017

Author’s response to reviews:

Dear Editor,

Dear Reviewers,

We have taken into account the feedback regarding the meaning of “pilot study”. We have eliminated the word “pilot” from the title.

We have maintained the focus throughout the manuscript that this study is the first of its type in the population targeted by the assessment.

The text formatting is in compliance with the format required (2 space, page numbers and line numbers).

Ethics approval and consent to participate, Competing interests, Funding, Authors’ Contributions and Acknowledgements are clearly stated.

Availability of data and materials: the database will be added and made available.

Consent to publish: as per journal regulations and article type - not applicable. No image, record, declaration or other types of material regulated by specific consent to publish from the participants were provided in the manuscript or used for the preparation of the manuscript.
The authors signed a license agreement as per journal regulations.

Thank you kindly for the insightful feedbacks!

Corresponding author